Author's response to reviews

Title: Practical knowledge of experienced nurses in critical care. A qualitative study of their narratives

Authors:

Maria Sagrario MS ACEBEDO-URDIALES (msagrario.acebedo@urv.cat)
José Luis JL MEDINA-MOYA (jlmedina@ub.edu)
Carme C FERRÉ-GRAU (carme.ferre@urv.cat)

Version: 5
Date: 14 July 2014

Author's response to reviews: see over
For the attention of the editorial board of BCM Medical Education

We are attaching a manuscript that we would like to submit for publication in the Journal Medical Education. We are a relatively new research group made up of experienced researchers from the Rovira i Virgili University and Barcelona University (Spain).

M. Sagrario ACEBEDO-URDIALES is a professor at the Universitat Rovira i Virgili in Tarragona. She has a doctorate in Nursing Sciences and a degree in Psychology and Nursing. She is head of the degree in Nursing. José Luis MEDINA-MOYA is a professor at the University of Barcelona. He has a doctorate in Pedagogy and Education Sciences and a degree in Nursing. Carmen FERRÉ-GRAU is a professor at the Universitat Rovira i Virgili in Tarragona. She has a doctorate in Psychology and a degree in Nursing. She is the director of the doctoral program in Nursing Sciences at the Universitat Rovira i Virgili in Tarragona.

We have decided to submit our article to Medical Education not only because of the prestige of the journal, but also because our article focuses on the practical knowledge of the ICU care of patients. This is an area that has received little attention from researchers and we therefore feel that it would be of interest to Medical Education readers.

Yours sincerely,

M. Sagrario Acebedo